Author's response to reviews

Title: GRAIDs: A framework for closing the gap in the availability of health promotion programs and interventions for people with disabilities

Authors:

James H Rimmer (jrimmer@uab.edu)
Kerri A Vanderbom (kerriv@lakeshore.org)
Linda Bandini (lbandini@bu.edu)
Charles Drum (Charles.drum@unh.edu)
Karen Luken (karenluken@gmail.com)
Yolanda Suarez-Balcazar (ysuarez@uic.edu)
Ian D Graham (igraham@ohri.ca)

Version: 6
Date: 21 July 2014

Author's response to reviews:

Please see attached cover letter.